

Hypothetical interventions and risk of atrial fibrillation by sex and education: application of the parametric g-formula in the Tromsø Study

Linn Nilsen ()¹*, Ekaterina Sharashova ()¹, Maja-Lisa Løchen ()¹, Goodarz Danaei ()^{2,3}, and Tom Wilsgaard ()¹

¹Department of Community Medicine, UiT The Arctic University of Norway, PO Box 6050 Langnes, N-9037 Tromsø, Norway; ²Department of Global Health and Population, Harvard TH Chan School of Public Health, Boston, MA, USA; and ³Department of Epidemiology, Harvard TH Chan School of Public Health, Boston, MA, USA

Received 2 February 2023; revised 3 July 2023; accepted 17 July 2023; online publish-ahead-of-print 19 July 2023

See the editorial comment for this article 'Habits and heartbeats: learning from historical longitudinal data on primary prevention of atrial fibrillation', by O. Sapir et al., https://doi.org/10.1093/eurjpc/zwad298.

Aims	To use the parametric g-formula to estimate the long-term risk of atrial fibrillation (AF) by sex and education under hypo- thetical interventions on six modifiable risk factors.		
Methods and results	We estimated the risk reduction under hypothetical risk reduction strategies for smoking, physical activity, alcohol intake, body mass index, systolic, and diastolic blood pressure in 14 923 women and men (baseline mean age 45.8 years in women and 47.8 years in men) from the population-based Tromsø Study with a maximum of 22 years of follow-up (1994–2016). The estimated risk of AF under no intervention was 6.15% in women and 13.0% in men. This cumulative risk was reduced by 41% (95% confidence interval 17%, 61%) in women and 14% (-7% , 30%) in men under joint interventions on all risk factors. The most effective intervention was lowering body mass index to $\leq 25 \text{ kg/m}^2$, leading to a 16% (4%, 25%) lower risk in women and a 14% (6% , 23%) lower risk in men. We found significant sex-differences in the relative risk reduction by sufficient physical activity, leading to a 7% (-4%, 18%) lower risk in women and an 8% (-2% , -13%) increased risk in men. We found no association between the level of education and differences in risk reduction by any of the interventions.		
Conclusion	The population burden of AF could be reduced by modifying lifestyle risk factors. Namely, these modifications could have prevented 41% of AF cases in women and 14% of AF cases in men in the municipality of Tromsø, Norway during a maximum 22-year follow-up period.		
Lay summary	The heart normally has a regular rhythm. However, in an increasing number of adults worldwide, the rhythm is irregular, which is known as arrhythmia. Atrial fibrillation, or AF, is the most common type of arrhythmia. We know that the risk of AF may be related to lifestyle. In this project, we investigated how much the risk of AF in the population could have been reduced by improvements in smoking habits, physical activity level, alcohol intake, body mass index (BMI), and blood pressure. We found that the risk could have been reduced by 41% in women and 14% in men if everyone quit smoking, was sufficiently physically active, limited their alcohol intake to two units per week, lowered their BMI to 25 kg/m ² , and lowered their blood pressure to 130/80 mm Hg. Reducing BMI was the most effective intervention to prevent AF.		

* Corresponding author. Tel: +47 77 64 63 52, Email: linn.nilsen@uit.no

© The Author(s) 2023. Published by Oxford University Press on behalf of the European Society of Cardiology.

This is an Open Access article distributed under the terms of the Creative Commons Attribution-NonCommercial License (https://creativecommons.org/licenses/by-nc/4.0/), which permits non-commercial re-use, distribution, and reproduction in any medium, provided the original work is properly cited. For commercial re-use, please contact journals.permissions@oup.com



Introduction

Atrial fibrillation (AF) is the most common cardiac arrhythmia and the cause is often unknown.¹ The risk of AF is associated with unmodifiable risk factors like genetics, sex, age, and ethnicity, and modifiable risk factors like hypertension, obesity, diabetes mellitus, obstructive sleep apnoea, physical activity, smoking, and alcohol consumption.^{1,2} Blood pressure (BP) is among the strongest modifiable risk factors in women, while body mass index (BMI) is in men.³ AF increases the risk of ischaemic stroke, myocardial infarction, heart failure, and premature death.² Worldwide prevalence in adults is estimated to be between 2% and 4%, and a 2.3-fold rise is expected.⁴ Additionally, the prevalence of modifiable risk factors such as hypertension, obesity, diabetes, and physical inactivity is increasing, and AF imposes a significant burden on both the patient and the health care system.⁵ Thus, preventing AF is a significant public health challenge.

Modification of lifestyle risk factors to prevent AF is now highlighted as a potential fourth pillar in AF management together with anticoagulation, rate control, and rhythm control.⁴ Additionally, individual changes in modifiable risk factors like systolic and diastolic BP (SBP and DBP) and BMI affect the incidence rate of AF and are important targets for primary prevention.^{3,6} Therefore, a risk modification strategy is crucial to reduce morbidity, years of life lost, and the medical costs attributable to AF. However, the literature has several gaps and specific AF primary prevention strategies including modification of lifestyle risk factors are few.^{2,4,7,8}

The current knowledge on modifiable risk factors and AF is mostly based on observational studies.² A few randomized controlled trials have examined the effects of interventions on lifestyle and AF risk or AF symptoms, showing inconclusive results.^{9–12} However, no study has yet examined the effect of joint lifestyle interventions for primary prevention of AF. For this objective, randomized controlled trials are the gold standard, but they are often time consuming not feasible, unaffordable, or unethical. Therefore, the emulation of a target trial by using Robins' parametric g-formula is an alternative method to assess the impact of interventions on the risk of incident AF.¹³ To our knowledge, only one previous study has applied the parametric g-formula to estimate the effect of interventions on AF risk, but that study focused only on BMI.⁶

Persons with low socioeconomic status (SES) have poorer cardiovascular health.¹⁴ The current literature on AF and SES is, however, inconclusive. A review from 2018 found no association between SES and incident AF,¹⁵ but more recent studies have found socioeconomic disadvantages across the life course and low family income to be associated with increased AF risk.^{16,17}

We aimed to estimate the effect of various hypothetical interventions using the parametric g-formula on lifestyle and metabolic risk



factors on the risk of incident AF and to investigate if the effect varied with sex and SES using data from the population based on the longitudinal Tromsø Study.

Materials and methods

Study population

Tromsø is the largest municipality in Northern Norway with 78 000 residents.¹⁸ Around 90% of the residents live in urban areas, and 85% were born in Norway.¹⁸ The Tromsø Study is a prospective, population-based cohort study conducted in the municipality of Tromsø, Norway. It consists of seven completed surveys, Tromsø1-7, conducted between 1974 and 2016.¹⁹ Tromsø4 (1994-1995) was used as baseline in this study, and all participants were followed up with for incident AF both prospectively and retrospectively. All residents of Tromsø aged ≥ 25 years were invited to Tromsø4, with 72% participation (n = 27158). Of these, n = 26878 gave written informed consent. Because we needed at least one pre-baseline measurement in order to adjust for pre-baseline covariates, only n = 16415Tromsø4 participants that also participated in Tromsø2 (1979–1980, n = 16621) or Tromsø3 (1986–1987, n = 21862) were eligible for this study. Of these, n = 9661 also attended Tromsø5 (2001, n =8130) and/or Tromsø6 (2007–2008, n = 12 984), where random samples or selected birth cohort were invited. Among those eligible for our study, 91% of those invited to Tromsø5, and 82% of those invited to Tromsø6, participated in Tromsø 5 and Tromsø 6, respectively. Participants were excluded from the analyses if they were pregnant at baseline (n = 116), had prevalent AF (n = 73), had moved out of the municipality before their date of examination (n = 10), had

incomplete covariate history pre-baseline (n = 963) or at baseline after carrying data forward for one survey (n = 330). Women who were pregnant at Tromsø2 or Tromsø3 had their covariates set to missing for that time point. In total, n = 14923 women and men were included (*Figure 1*).

Measurements

Data collection in the Tromsø Study included questionnaires, physical examinations, and blood sampling. The different surveys used standardized protocols and similar methods that are described elsewhere.¹⁹ In brief, information on education, smoking, physical activity during leisure time, alcohol consumption, diabetes, history of heart attack or stroke, marital status, physical activity at work, and pregnancy was collected from questionnaires. Detailed information measuring smoking and harmonization of education, physical activity during leisure time, and alcohol consumption across the different surveys is given in the *Supplementary material online*. Height, weight, and BP were measured, and non-fasting blood samples were collected by trained personnel at the physical examination (*Supplementary material online*). Emigration and death were identified through the Population Register of Norway.

Identification and validation of incident AF

Incident cases of AF were identified by linkage to the diagnosis registry at the University Hospital of North Norway and the Norwegian Cause of Death Registry using the unique Norwegian national 11-digit identification number. Both in- and out-patient clinical diagnoses are included in the registry. Potential incident cases of AF were detected for validation by manual and/or digital searches for the International Classification of Diseases, 9th Revision



Figure 2 Directed acyclic graph of the causal structure between time-varying exposures, time-varying covariates, and the risk of atrial fibrillation.

(ICD-9) codes 410–414, 427, 428, 430–438, and 798–799, and ICD-10 codes 120–125, 146–148, 150, 160–169, R96, R98, and R99.²⁰ Additionally, for participants with cerebrovascular or cardio-vascular events but without an arrhythmia diagnosis, medical hospital records were searched for notes on AF.²¹ An independent endpoint committee confirmed and validated all AF events documented on an electrocardiogram following a strict protocol.²¹ Participants with suspected AF but without documentation on an electrocardiogram (n = 105), those having AF after the end of follow-up period (n = 65), and those having AF after moving out of the municipality (n = 24) were considered AF-free in the analyses. One participant had a date of AF after a date of death. The date of AF was in this case changed to match the date of death.

Follow-up and missingness

Each participant was followed up with from the date of participation in Tromsø4 until the date of first documented AF, emigration, death, or end of follow-up period (31 December 2016), whichever came first. The maximum follow-up period was 22 years. AF risk factors were updated for participants of Tromsø5 or Tromsø6 that were still in the follow-up period. The last observation was carried forward from the previous survey if a covariate was missing for one of the time points. Thus, in accordance with exclusion criteria, all participants had complete covariate history pre-baseline (e.g. Tromsø2 and/or 3) and baseline (e.g. Tromsø4). Participants were censored on 31 December 2008 after carrying data over from Tromsø4 to Tromsø5 if a covariate was missing for both Tromsø5 and Tromsø6.

Risk reduction strategies

We conducted hypothetical interventions on six modifiable risk factors, chosen based on reviews and clinical guidelines.^{2,4} In our model, we made all participants hypothetically (i) quit smoking, (ii) become sufficiently physically active (i.e. perform at least 180 min per week of moderate physical activity or 90 min per week of vigorous physical activity), (iii) drink alcohol moderately (>one \leq two units per week), (iv) lower their BMI to 25 kg/m², (v) lower their SBP to 130 mm Hg, and (iv) lower their DBP to 80 mm Hg. These interventions were performed at each time point, both individually and as combinations of interventions 5 + 6, 1–4, and all joint interventions 1–6.

Statistical methods

All statistical analyses were performed using SAS software version 9.4 (SAS Institute, Cary, NC, USA) and stratified by sex, due to known sexdifferences in AF epidemiology.^{3,4,20,22–27} Survey-specific descriptive characteristics of the study sample were estimated by means (standard deviations) and proportions (numbers).

We used the parametric g-formula to estimate the long-term risk of AF under no intervention and under the selected interventions described above. The parametric g-formula is a generalization of standardization for time-varying exposures, and AF risk is estimated by a weighted sum of overall risk factor histories of the probability of AF conditional on its risk factors.²⁸ The parametric g-formula also correctly adjusts for time-varying confounders, which is a methodological challenge where standard regression models fail. For example, to assess the effect of interventions on BMI on AF risk, physical activity is a timevarying confounder because it affects the risk of AF, and changes in physical activity may lead to changes in future BMI. Additionally, physical activity level is affected by past BMI. The most common adjustment method is to add both BMI and physical activity as time-varying covariates in a regression model. This allows us to estimate the effect of present BMI on AF risk, but not the total effect of past and present or future BMI (and hence the effect of change in BMI). This total effect can be biased because adjusting for present physical activity is equal to conditioning on a collider (between past physical activity and past BMI), and this may introduce selection bias.^{13,29} Robins' parametric g-formula can overcome this bias. The parametric g-formula has previously been used to estimate the effect of hypothetical interventions on, among others, the risk of myocardial infarction, 30,31 ischaemic stroke,^{32,33} and diabetes.³

In short, the estimation process is as follows. First, for each timeperiod from the baseline to the end of the follow-up period, use linear and logistic regression to model each risk factor, risk of non-AF death, and risk of AF as a function of prior risk factor history. Second, simulate a cohort under the selected interventions in five steps: (i) use the observed values of covariates at baseline; (ii) predict values of covariates at the next time point using the coefficients from the regression models; (iii) 'intervene' by setting the values of the covariates to the intervention-values; (iv) predict the probability of AF and non-AF death using these new values; (iv) repeat steps (ii) through (iv) for each time period and estimate the population risk as the average of the subjectspecific risks. The 95% confidence limits were defined as the 2.5% and 97.5%iles using non-parametric bootstrapping with 1000 samples.

Time-fixed variables included in the model as potential confounders were baseline (e.g. Tromsø4 covariates) sex, age, education, marital status, physical activity at work, history of myocardial infarction and stroke, and pre-baseline (e.g. Tromsø2 and Tromsø3 covariates) smoking status, physical activity during leisure time, alcohol consumption, BMI, SBP, DBP, and total cholesterol level. The number of cigarettes per day, physical activity during leisure time, alcohol consumption, BMI, diabetes mellitus, SBP, DBP, total cholesterol, and HDL cholesterol were modelled as time-varying covariates in the listed order. In *Figure 2*, we present a directed acyclic graph (DAG) of the causal structure between time-varying exposures, time-varying covariates, and the risk of AF. In Supplementary material online, *Figure S2*, we present a DAG of the assumed causal relationship between variable measurements at each visit and AF. Both DAGs are made using DAGitty.³⁵ We present population risk ratios and risk differences by comparing the estimated long-term risk of AF under each intervention with the risk under no intervention (the natural course), in addition to the average and cumulative percent intervened. The validity of our models was examined by comparing the observed risk of AF and death, and the observed means of the time-varying confounders, with those predicted by the models.

We investigated if the effect of the interventions varied by education level by performing a sub-group analysis on those with university/college education (\geq 4 years and <4 years) vs. lower levels (high school 10–12 years and/or primary school 7–10 years). Effect modification was also assessed by sub-groups of sex. The SAS macro and its documentation are available online (https://causalab.sph.harvard.edu/software/).

Results

Baseline characteristics

We included 14 923 women and men aged 28–82 years (*Table 1*). For women, mean age at baseline was 46 years, 26% had college or university-level education, 40% were daily smokers, 49% were sufficiently physically active, 63% had an alcohol intake of at least one unit per week, mean (standard deviation) BMI was 24.7 (4.0) kg/m², and SBP was 129 (18.6) mmHg. For men, the mean age at baseline was 48 years, 29% had college or university-level education, 37% were daily smokers, 55% were sufficiently physically active, 78% had an alcohol Intake of at least one unit per week, mean BMI was 25.8 (3.3) kg/m², and SBP was 137 (16.8) mmHg. In women and men, respectively, there were 420 and 932 incident cases of AF and 588 and 1130 deaths during a maximum 22-year follow-up period. The incidence rate of AF was 2.92 per 1000 person-years in women and 6.87 per 1000 person-years in men.

Simulated and observed risk of AF

The simulated and observed long-term risk of AF was 6.15% and 7.04% in women, and 13.0% and 14.6% in men (*Table 2*). For the time-varying covariates, the simulated and observed values had small mean differences, indicating that the model under the null was satisfactory (see Supplementary material online, *Figure S1*).

Effect of single interventions

Of the interventions, only the reduction of BMI was statistically significant, associated with a 16% reduced risk in women [95% confidence interval (CI) 4%, 25%], and a 14% reduced risk in men (95% CI 6%, 23%) (*Table 2*). Although not significant, all other single interventions were associated with a reduced risk of AF in women. In men, weaker intervention effects were observed, and interventions to become sufficiently physically active were significantly associated with an 8% increased risk.

Effect of joint interventions

The joint intervention on smoking, physical activity, alcohol, and BMI was significantly associated with a reduced risk of 35% in women (95% CI 9%, 54%). The joint intervention on all covariates (smoking, physical activity, alcohol, BMI, SBP, and DBP) was associated with a significantly reduced risk of 41% in women (95% CI 17%, 61%). In men, the

joint interventions were protective but not significant. The average percent intervened on was 87% in women and 94% in men for the joint intervention on all covariates.

Sub-group analysis by sex and education level

In sub-group analyses of women and men, the relative and absolute effect of being sufficiently physically active was significantly different between women and men [There was a 7% reduced risk (95% CI –4%, 18%) in women vs. an 8% increased risk (95% CI –2%, -13%) in men, and a risk difference of -0.43 (-1.10, 0.25) in women and of 1.01 (0.23, 1.78) in men] (*Table 2*). We observed a borderline significant sex difference for the joint intervention on all covariates, P = 0.06. In sub-groups analyses of education at university/college level compared to high school and/or primary school, no significant differences in the relative or absolute effect of any interventions were found (*Table 3*).

Discussion

We found that risk reduction strategies including quitting smoking, sufficient physical activity, moderate alcohol intake, BMI reduction, and lowering BP could have prevented 41% of incident AF in women and 14% in men. We found notable sex differences in the relative risk reductions by joint interventions and in the relative and absolute risk reduction by sufficient physical activity. We found no significant differences in the absolute or relative effect of any interventions between the educational level sub-groups.

In our study, lowering BMI was the only intervention that was significantly associated with reduced risk of AF in both women and men. In men, we found a slightly increased risk of AF by the intervention of physical activity. Some studies have demonstrated a high prevalence of AF in male endurance athletes, and studies investigating wider ranges of physical activity levels and risk of AF have found results varying from decreased risk to a U-shaped relationship, but increased risk has also been demonstrated.³⁶ The Tromsø Study has previously shown a J-shaped association between physical activity and AF with an increased risk of AF in the highest physical activity levels, especially among men.³⁶ This may explain parts of the finding in our study of an increased risk of AF by the intervention in physical activity in men. This may also be a part of the explanation for the difference in the effect of the joint interventions between women and men (41% vs. 14% reduced risk) because the effect of the intervention on physical activity may outweigh the beneficial effect of reducing BMI in men, but not in women. Our study found small effects of each single intervention (3-15% change in risk). However, the effect of the joint interventions was greater overall (41% reduced risk in women and 14% reduced risk in men). This may imply multiplicative effects where several small changes have large benefits regarding the risk of AF, especially in women.

If existing literature demonstrated a clear social gradient in AF risk, one could also expect to find a social gradient in the effect of lifestyle interventions on AF risk. However, existing literature on these risks is inconclusive. A systematic review from 2018 that includes 12 studies found no association between education and the risk of AF.¹⁵ A Danish study from 2020 that includes almost 2.5 million individuals found that a higher level of education was associated with a lower risk of AF in young individuals, but this association decreased with age and was almost absent for the oldest age groups.³⁷ We did not find any significant differences in the absolute or relative effect of the joint interventions in the different levels of education. To our knowledge, no other studies have investigated the effect of interventions on modifiable risk factors on AF risk in levels of SES or investigated if the effect of modifiable risk factors on AF risk differs between levels of SES.

Table 1 Descriptive characteristics by sex and survey^a

	Women			Men		
	Tromsø4 1994–95	Tromsø5 2001	Tromsø6 2007–08	Tromsø4 1994–95	Tromsø5 2001	Tromsø6 2007–08
n	7418	2633	3888	7505	2381	3671
Age, years	45.8 (9.3)	52.7 (9.2)	60.8 (9.3)	47.8 (10.6)	54.3 (10.3)	61.7 (10.0)
Education						
≤10 years of schooling	37.0 (2744)	37.2 (2549)	38.6 (1493)	32.4 (2434)	32.0 (2151)	32.0 (1153)
High school diploma	37.3 (2764)	37.5 (2574)	36.3 (1405)	38.3 (2878)	38.6 (2595)	38.6 (1392)
College or university <4 years	13.2 (981)	13.1 (897)	12.5 (483)	16.4 (1233)	16.8 (1128)	17.4 (628)
College or university \geq 4 years	12.5 (929)	12.3 (841)	12.6 (486)	12.8 (960)	12.6 (843)	11.9 (430)
Daily smoking, %						
Non-smoker	59.9 (4444)	61.8 (4241)	78.7 (3045)	62.6 (4696)	64.6 (4340)	81.3 (2929)
1–4 cigarettes/day	3.0 (224)	2.8 (195)	1.4 (55)	1.8 (137)	1.9 (126)	0.9 (34)
5–14 cigarettes/day	27.3 (2023)	26.4 (1812)	14.9 (575)	18.4 (1380)	17.9 (1203)	11.4 (410)
15–24 cigarettes/day	9.0 (668)	8.2 (566)	4.6 (178)	14.5 (1087)	13.5 (905)	5.9 (211)
≥25 cigarettes/day	0.8 (59)	0.7 (47)	0.4 (14)	2.7 (205)	2.1 (143)	0.5 (19)
Leisure time physical activity ^b , %						
Inactive	7.0 (517)	6.0 (412)	4.3 (166)	7.4 (557)	6.3 (425)	6.1 (221)
Insufficiently active	43.7 (3239)	42.2 (2897)	50.6 (1957)	37.6 (2823)	37.2 (2496)	52.8 (1902)
Sufficiently active	49.4 (3662)	51.8 (3552)	45.1 (1744)	55.0 (4125)	56.5 (3796)	41.1 (1480)
Alcohol consumption, %						
0 units per week	37.2 (2758)	34.1 (2343)	16.2 (626)	22.0 (1652)	19.8 (1333)	9.1 (329)
$>0 \le 1$ unit per week	23.0 (1703)	23.2 (1590)	29.0 (1120)	15.4 (1159)	15.1 (1015)	23.0 (828)
$>1 \leq 2$ units per week	16.9 (1253)	17.4 (1191)	24.4 (942)	15.7 (1179)	16.0 (1073)	19.8 (714)
$>2 \leq 3$ units per week	9.3 (691)	10.0 (688)	8.4 (326)	12.8 (958)	13.2 (889)	15.5 (557)
$>3 \le 4$ units per week	5.3 (390)	5.7 (393)	13.2 (510)	9.2 (689)	9.1 (611)	15.4 (556)
>4 units per week	8.4 (623)	9.6 (656)	8.9 (343)	24.9 (1868)	26.7 (1796)	17.2 (619)
Body mass index, kg/m ²	24.7 (4.0)	25.2 (4.3)	26.7 (4.6)	25.8 (3.3)	26.0 (3.4)	27.2 (3.6)
Systolic blood pressure, mm Hg	129 (18.6)	130 (19.2)	137 (24.1)	137 (16.8)	137 (17.0)	141 (20.7)
Total serum cholesterol, mmol/L	6.0 (1.3)	6.0 (1.2)	5.9 (1.1)	6.2 (1.2)	6.0 (1.2)	5.6 (1.1)
Serum high-density lipoprotein	1.7 (0.4)	1.6 (0.4)	1.7 (0.4)	1.4 (0.4)	1.4 (0.3)	1.4 (0.4)
cholesterol, mmol/L						
Diabetes, %	0.9 (64)	1.4 (95)	4.8 (184)	1.6 (117)	1.9 (125)	5.5 (198)
Heart attack at baseline, %	0.5 (38)			3.7 (275)		
Stroke at baseline, %	0.6 (45)			1.2 (92)		
Marital status						
Single	17.4 (1293)	17.3 (1184)	13.9 (536)	22.1 (1656)	22.3 (1499)	18.0 (648)
Married/registered partnership	64.6 (4789)	65.3 (4480)	68.7 (2655)	65.6 (4925)	66.1 (4443)	71.0 (2558)
Widow/widower	4.0 (298)	3.9 (266)	4.5 (174)	1.4 (103)	1.2 (81)	1.2 (45)
Divorced	11.8 (872)	11.5 (790)	11.2 (433)	9.1 (680)	8.6 (579)	8.2 (295)
Separated	2.2 (166)	2.1 (141)	1.8 (69)	1.9 (141)	1.7 (115)	1.6 (57)
Physical activity at work						
Mostly sedentary	40.4 (2996)	40.3 (2768)	40.3 (1559)	46.4 (3479)	46.0 (3091)	46.8 (1688)
A lot of walking	36.2 (2683)	36.3 (2489)	36.3 (1404)	24.1 (1811)	24.4 (1640)	25.1 (905)
A lot of walking and lifting	21.6 (1601)	21.5 (1474)	21.1 (815)	18.8 (1413)	19.1 (1282)	17.9 (645)
Heavy manual labour	1.9 (138)	1.9 (130)	2.3 (89)	10.7 (802)	10.5 (704)	10.1 (365)

The Tromsø Study 1994–2008.

^aNumbers are given as percent (number) or as mean (standard deviation).

^bInactive = no minutes of light or hard physical activity per week. Sufficiently active ≥180 min per week moderate physical activity or ≥90 min: insufficiently active = all other levels.

To our knowledge, only one previous study has investigated the effect of interventions on incident AF risk using the parametric g-formula.⁶ Conner et al. only considered BMI as an intervention variable and found a 30% (95% Cl 2%, 50%) risk reduction in women and an

18% (95% Cl -1%, 34%) risk reduction in men for BMI 18.5–29.9 kg/m² compared to BMI 30–41 kg/m^{2.6} This is in line with our main finding of BMI as the only single intervention of statistical significance in both women and men. However, because results from the parametric

The Tromsø Study 1994–2008.

AF, atrial fibrillation; BMI, body mass index; CI, confidence interval; DBP, diastolic blood pressure; SBP, systolic blood pressure.

^aEstimated using the parametric g-formula with fixed covariates: age, sex, education, former smoking, marital status, work time physical activity and history of myocardial infarction and/or stroke; and time-varying covariates smoking, physical activity, alcohol use, BMI, systolic and diastolic blood pressure, total cholesterol and diabetes mellitus.

^bObserved risk 7.04% in women and 14.59% in men.

^cThe percentage of the population intervened in at least one of the periods.

^dAverage percent of the population intervened in a given period.

*Test for equality between men and women, 0.05 < P < 0.10.

**Test for equality between men and women, P < 0.05.

g-formula are strongly dependent on baseline exposure values, the absolute and relative risk reductions are not always comparable across studies. In this case, baseline BMI was higher in the study from Conner et al. than in our study. Additionally, the target BMIs for interventions were different.

Few other studies have reported the effects of prospective risk factor reduction on AF primary prevention.² The Tromsø Study has recently demonstrated that individual changes in SBP and DBP in women and in BMI in men had the largest contribution among the studied risk factors on changes in AF incidence.³ Reviews have summarized some of the existing, mostly observational, evidence on lifestyle and risk factor modification for reduction of AF incidence.^{1,2,38} The reviewed studies have, however, used more conventional methods that require additional assumptions and can, therefore, not be directly compared with our estimates.

This study has several strengths. First, the Tromsø Study is a large, population-based cohort with high participation proportions, and includes participants from both urban and rural areas. The study sample is similar to the general Norwegian adult population regarding age and sex, but physical activity level and educational level are slightly higher than in the general population.³⁹ Consequently, a limitation of this study may be an overrepresentation of physically active persons and persons with higher education. Second, AF incidence is ascertained with high sensitivity, and events from both in- and out-patient clinics are included from the only hospital in the area. However, up to 40% of AF patients are asymptomatic, and some symptomatic cases are treated in primary health care only and never referred to the hospital. Therefore, a likely limitation is that the incidence is underestimated because not all AF patients seek the hospital.³ Third, we adjusted for time-varying confounders affected by prior exposures and simulated long-term joint interventions on modifiable risk factors using the parametric g-formula. Time-varying confounding is a methodological challenge where many conventional methods fail.²⁹ Methods that use stratification, regression,

Table 2 F	Risk of atrial fibrillation	under hypothetical	interventions by sex ^a
-----------	-----------------------------	--------------------	-----------------------------------

No.	Intervention	22-year risk of AF, % (95% CI)	Population risk ratio (95% Cl)	Population risk difference ^b (95% CI)	Cumulative percent intervened on ^c	Average per cent intervened on ^d
	Women (<i>n</i> = 7418)					
0	Natural course	6.15 (5.47, 6.78)	1	0	0	0
1	All become non-smokers	5.88 (5.02, 6.68)	0.96 (0.88, 1.04)	-0.26 (-0.71, 0.21)	41	15
2	Physical activity > 180 min low/moderate intensity or >90 min hard int	5.73 (4.89, 6.66)	0.93 (0.82, 1.04)**	-0.43 (-1.10, 0.25)**	62	27
3	Alcohol intake $>1 \le 2$ units per week	5.33 (3.97, 6.68)	0.87 (0.66, 1.08)	-0.82 (-2.05, 0.50)	90	47
4	$BMI \le 25 \text{ kg/m}^2$	5.19 (4.35, 6.03)	0.84 (0.75, 0.96)	-0.96 (-1.53, -0.23)	60	40
5	SBP ≤ 130 mm Hg	5.74 (4.63, 6.75)	0.93 (0.79, 1.08)	-0.41 (-1.28, 0.47)	67	43
6	DBP ≤ 80 mmHg	6.02 (5.23, 6.78)	0.98 (0.90, 1.05)	-0.13 (-0.61, 0.31)	53	29
7	Interventions 5–6	5.55 (4.67, 6.41)	0.90 (0.79, 1.02)	-0.60 (-1.31, 0.11)	77	54
8	Interventions 1–4	4.02 (2.79, 5.50)	0.65 (0.46, 0.91)*	-2.13 (-3.40, -0.53)	99	73
9	Interventions 1–6	3.61 (2.37, 5.04)	0.59 (0.39, 0.83)*	-2.54 (-3.74, -1.07)	100	87
	Men (<i>n</i> = 7505)					
0	Natural course	13.0 (12.1, 14.0)	1	0	0	0
1	All become non-smokers	13.4 (12.3, 14.7)	1.03 (0.99, 1.08)	0.38 (-0.15, 1.02)	40	14
2	Physical activity > 180 min low/moderate intensity or >90 min hard int	14.0 (12.6, 15.4)	1.08 (1.02, 1.13)**	1.01 (0.23, 1.78)**	58	24
3	Alcohol intake $>1 \le 2$ units per week	12.2 (10.3, 14.2)	0.94 (0.81, 1.09)	-0.82 (-2.49, 1.15)	92	49
4	$BMI \le 25 \text{ kg/m}^2$	11.1 (9.79, 12.6)	0.86 (0.77, 0.94)	-1.89 (-2.95, -0.80)	74	53
5	SBP ≤ 130 mm Hg	12.1 (10.6, 13.5)	0.93 (0.84, 1.01)	-0.97 (-2.07, 0.18)	84	56
6	$DBP \le 80 \text{ mmHg}$	13.6 (12.4, 15.1)	1.05 (0.99, 1.12)	0.61 (-0.18, 1.56)	72	43
7	Interventions 5–6	12.7 (11.4, 14.0)	0.97 (0.89, 1.05)*	-0.35 (-1.38, 0.63)	91	70
8	Interventions 1-4	11.5 (9.35, 14.1)	0.89 (0.73, 1.08)*	-1.50 (-3.53, 1.00)	99	80
9	Interventions 1–6 ⁹	11.3 (8.91, 14.1)	0.86 (0.70, 1.07)*	-1.77 (-3.99, 0.93)	100	94

No.	Intervention	Population risk ratio (9	5% CI)	Population risk difference ^b		
		Education at university/college level	High school and/or primary school	Education at university/ college level	High school and/or primary school	
0	Natural course	Ref. (8.94% risk)	Ref. (9.74%)	Ref. (8.94% risk)	Ref. (9.74%)	
1	All become non-smokers	1.02 (0.95, 1.07)	1.02 (0.97, 1.07)	0.17 (-0.47, 0.61)	0.15 (-0.34, 0.67)	
2	Physical activity > 180 min low/moderate intensity or >90 min hard int	1.03 (0.92, 1.11)	1.04 (0.98, 1.11)	0.31 (-0.70, 1.01)	0.42 (-0.20, 1.07)	
3	Alcohol intake $> 1 \le 2$ units per week	0.85 (0.66, 1.07)	0.93 (0.81, 1.06)	-1.38 (-3.18, 0.64)	-0.66 (-1.91, 0.62)	
4	BMI \leq 25 kg/m ²	0.91 (0.76, 1.05)	0.85 (0.78, 0.93)	-0.83 (-2.14, 0.40)	-1.47 (-2.19, -0.73)	
5	SBP ≤ 130 mm Hg	0.85 (0.70, 0.98)	0.97 (0.87, 1.05)	-1.35 (-2.75, -0.22)	-0.33 (-1.26, 0.48)	
6	DBP ≤ 80 mmHg	1.10 (1.01, 1.20)	1.01 (0.95, 1.07)	0.87 (0.08, 1.84)	0.06 (-0.49, 0.65)	
7	Interventions 5–6	0.94 (0.80, 1.06)	0.97 (0.89, 1.04)	-0.50 (-1.82, 0.52)	-0.30 (-1.08, 0.34)	
8	Interventions 1-4	0.78 (0.55, 1.06)	0.84 (0.69, 1.01)	-1.93 (-4.06, 0.57)	-1.55 (-3.07, 0.07)	
9	Interventions 1–6	0.72 (0.48, 1.02)	0.82 (0.65, 0.98)	-2.54 (-4.81, 0.14)	-1.79 (-3.44, -0.20)	

 Table 3
 Risk of atrial fibrillation under hypothetical interventions by education at baseline^a

The Tromsø Study 1994–2008.

BMI, body mass index; CI, confidence interval; DBP, diastolic blood pressure; SBP, systolic blood pressure.

^aEstimated using the parametric g-formula with fixed covariates: age, sex, education, former smoking, marital status, work time physical activity and history of myocardial infarction and/or stroke; and time-varying covariates smoking, physical activity, alcohol use, BMI, systolic and diastolic blood pressure, total cholesterol and diabetes mellitus. Test for equality between sub-groups of education at university/college level (\geq 4 years and <4 years) and high school (10–12 years) and/or primary school (7–10 years), found no significant differences (all P > 0.05).

^bObserved risk 10.25% for participants with education at university/college level and 11.09% for participants with high school and/or primary school.

or matching to adjust for time-varying covariates will fail to estimate the joint effect of a time-varying covariate on an outcome because these methods for adjustment introduce selection bias.²⁹ Thus, by eliminating time-varying confounding they introduce a new bias. The parametric g-formula overcomes this bias and provides unbiased and properly adjusted effects of time-varying covariates affected by prior exposures.²⁹

Our results are only valid under the general assumptions for cohort studies of no model misspecification, no unmeasured or residual confounding, and no measurement error. Our results support the absence of model misspecification under the null, as the observed and estimated risks under no intervention were rather similar (6.1% and 7.0% in women, and 13.0% and 14.6% in men). The estimated effects are increasingly model dependent when the average percent intervened on approaches 100% and, consequently, more prone to misspecification. This is a potential limitation of our study. Another potential limitation is that unmeasured confounding is plausibly present, as we adjusted for several confounders but did not include data on other potential confounders, for example, diet. Similarly, some measurement error is expected, especially when using self-reported variables on lifestyles like physical activity, smoking, and alcohol consumption. Physical activity and alcohol consumption had to be harmonized from different questionnaires across the included surveys, and, thus, are especially prone to information bias. Further, the parametric g-formula has a set of specific assumptions: counterfactual consistency, sequential exchangeability, and positivity, which are described in detail elsewhere.²⁹ Consistency implies that interventions should be well defined and that the counterfactual outcome under each intervention should be the same as the observed outcome under the same level of risk factor.^{29,32} As Vangen-Lønne et al. pointed out in a similar project on interventions for stroke incidence, the consistency assumption may hold for lifestyle and behavioural risk factors such as smoking and alcohol use but is less likely to hold for metabolic risk factors such as BMI and BP.³² Consequently, the estimated effects should be interpreted as the effect of a combination of changes or interventions

that led to a reduction in BMI or BP observed in the study population during the follow-up period.³² The sequential exchangeability assumption implies no uncontrolled confounding and no selection bias. The positivity assumption implies that there should be exposed and unexposed individuals within all confounder and prior exposure levels, i.e. that all observed treatment levels should be observed within all confounders. The parametric g-formula is also subject to the g-null paradox; the paradox that some model misspecification is guaranteed in some settings (e.g. when the null hypothesis is true) and as such the null hypothesis may be falsely rejected.²⁹ We attempted to avoid this paradox by keeping our model flexible and only considering interventions for which we in advance believed to have an effect. Another potential limitation is that 41% of our total cohort only attended the baseline and pre-baseline visits. However, we have no reason to suspect a systematic difference between those who were invited to later surveys and those who were not. Those who were invited to later surveys were on average older but were otherwise considered to be randomly sampled.

Conclusion

We found that the population burden of AF could be reduced by modifying lifestyle risk factors with hypothetical interventions in line with clinical guidelines. Two out of five incident cases of AF in women and almost one out of six cases in men that occurred during a maximum 22-year follow-up period in the Tromsø Study population could have been prevented by six hypothetical scenarios of intervention on lifestyle risk factors. The lowering of BMI was the most effective hypothetical intervention in both women and men. The effect of joint intervention did not differ significantly between those with college/ university level education compared to those with high school and/ or primary school studies, either on a relative scale or on an absolute scale.

Supplementary material

Supplementary material is available at European Journal of Preventive Cardiology.

Author contributions

All authors contributed to conceptualization, methodology, interpretation of data, and writing (review and editing) in this study. L.N. and T.W. did the formal analysis. M.L.L. contributed to the data collection. E.S., M.L.L., and T.W. contributed to supervision. L.N. did the visualization of data and results and the writing of the original draft.

Acknowledgements

We wish to extend our gratitude to all participants of the Tromsø Study, contributing to valuable knowledge and the progress of important research. The Tromsø Study has been supported by several sources, among others the Ministry of Health and Care Services, the Northern Norway Regional Health Authority, the University Hospital of North Norway, Troms County, and UiT the Arctic University of Norway. We also wish to thank the Publication Fund of UiT the Arctic University of Norway for their funding.

Funding

UiT the Arctic University of Norway funded this project and L.N.'s PhD project fellowship. The publication charges for this article have been funded by a grant from the Publication Fund of UiT the Arctic University of Norway.

Conflict of interest: M.L.L. has received lecture fees from Bayer, Sanofi, and BMS/Pfizer, not related to this study. The other authors declare no conflict of interest.

Data availability

The data underlying this article were provided by the Tromsø Study by permission. Data from the Tromsø Study are available upon reasonable request and application. More information can be found at http://www. tromsoundersøkelsen.no.

Previous presentations: These results have been orally presented at two epidemiology conferences: the NordicEPI conference in Reykjavik, Iceland, 18 August 2022, and at the Norwegian Epidemiological Association conference in Tromsø, Norway, 27 October 2022.

References

- Staerk L, Sherer JA, Ko D, Benjamin EJ, Helm RH. Atrial fibrillation. Circ Res 2017;120: 1501–1517.
- Chung MK, Eckhardt LL, Chen LY, Ahmed HM, Gopinathannair R, Joglar JA, et al. Lifestyle and risk factor modification for reduction of atrial fibrillation: A scientific statement from the American heart association. *Circulation* 2020;**141**:e750–e772.
- Sharashova E, Gerdts E, Ball J, Espnes H, Jacobsen BK, Kildal S, et al. Sex-specific time trends in incident atrial fibrillation and the contribution of risk factors: the Tromsø study 1994–2016. Eur J Prev Cardiol 2023;30:72–81.
- 4. Hindricks G, Potpara T, Dagres N, Arbelo E, Bax JJ, Blomström-Lundqvist C, et al. 2020 ESC guidelines for the diagnosis and management of atrial fibrillation developed in collaboration with the European Association for Cardio-Thoracic Surgery (EACTS): the task force for the diagnosis and management of atrial fibrillation of the European Society of Cardiology (ESC) developed with the special contribution of the European Heart Rhythm Association (EHRA) of the ESC. Eur Heart J 2021;42:373–498.
- Roth GA, Mensah GA, Johnson CO, Addolorato G, Ammirati E, Baddour LM, et al. Global burden of cardiovascular diseases and risk factors, 1990–2019: update from the GBD 2019 study. J Am Coll Cardiol 2020;76:2982–3021.

- Conner SC, Lodi S, Lunetta KL, Casas JP, Lubitz SA, Ellinor PT, et al. Refining the association between body mass Index and atrial fibrillation:gG-formula and restricted mean survival times. J Am Heart Assoc 2019;8:e013011.
- Kirchhof P, Breithardt G, Bax J, Benninger G, Blomstrom-Lundqvist C, Boriani G, et al. A roadmap to improve the quality of atrial fibrillation management: proceedings from the fifth atrial fibrillation network/European heart rhythm association consensus conference. Europace 2016;18:37–50.
- Menezes AR, Lavie CJ, DiNicolantonio JJ, O'Keefe J, Morin DP, Khatib S, et al. Atrial fibrillation in the 21st century: A current understanding of risk factors and primary prevention strategies. *Mayo Clin Proc* 2013;88:394–409.
- Albert CM, Cook NR, Pester J, Moorthy MV, Ridge C, Danik JS, et al. Effect of Marine Omega-3 fatty acid and vitamin D supplementation on incident atrial fibrillation: A randomized clinical trial. JAMA 2021;325:1061–1073.
- Alonso A, Bahnson JL, Gaussoin SA, Bertoni AG, Johnson KC, Lewis CE, et al. Effect of an intensive lifestyle intervention on atrial fibrillation risk in individuals with type 2 diabetes: the Look AHEAD randomized trial. Am Heart J 2015;**170**:770–777.e775.
- Abed HS, Wittert GA, Leong DP, Shirazi MG, Bahrami B, Middeldorp ME, et al. Effect of weight reduction and cardiometabolic risk factor management on symptom burden and severity in patients with atrial fibrillation: a randomized clinical trial. JAMA 2013;310: 2050–2060.
- Acharya P, Safarova MS, Dalia T, Bharati R, Ranka S, Vindhyal M, et al. Effects of vitamin D supplementation and 25-hydroxyvitamin D levels on the risk of atrial fibrillation. *American Journal of Cardiology* 2022;**173**:56–63.
- Robins J. A new approach to causal inference in mortality studies with a sustained exposure period—application to control of the healthy worker survivor effect. *Mathematical Modelling* 1986;**7**:1393–1512.
- Schultz WM, Kelli HM, Lisko JC, Varghese T, Shen J, Sandesara P, et al. Socioeconomic Status and cardiovascular outcomes. *Circulation* 2018;**137**:2166–2178.
- Lunde ED, Nielsen PB, Riahi S, Larsen TB, Lip GYH, Fonager K, et al. Associations between socioeconomic status, atrial fibrillation, and outcomes: a systematic review. *Expert Rev Cardiovasc Ther* 2018;**16**:857–873.
- Misialek JR, Rose KM, Everson-Rose SA, Soliman EZ, Clark CJ, Lopez FL, et al. Socioeconomic status and the incidence of atrial fibrillation in whites and blacks: the Atherosclerosis Risk in Communities (ARIC) study. J Am Heart Assoc 2014;3:e001159.
- Bonaccio M, Di Castelnuovo A, Costanzo S, De Curtis A, Persichillo M, Cerletti C, et al. Life course socioeconomic Status and risk of hospitalization for heart failure or atrial fibrillation in the Moli-Sani study cohort. Am J Epidemiol 2021;**190**:1561–1571.
- Tromsø Kommune. Fakta om Tromsø. https://tromso.kommune.no/fakta-om-tromso (20 September 2022 2019)
- Jacobsen BK, Eggen AE, Mathiesen EB, Wilsgaard T, Njølstad I. Cohort profile: the Tromsø study. Int J Epidemiol 2012;41:961–967.
- Sharashova E, Wilsgaard T, Ball J, Morseth B, Gerdts E, Hopstock LA, et al. Long-term blood pressure trajectories and incident atrial fibrillation in women and men: the Tromsø study. Eur Heart J 2020;41:1554–1562.
- Nyrnes A, Mathiesen EB, Njølstad I, Wilsgaard T, Løchen ML. Palpitations are predictive of future atrial fibrillation. An 11-year follow-up of 22,815 men and women: the Tromsø study. Eur J Prev Cardiol 2013;20:729–736.
- Espnes H, Ball J, Løchen ML, Wilsgaard T, Njølstad I, Mathiesen EB, et al. Sex-Specific associations between blood pressure and risk of atrial fibrillation subtypes in the Tromsø study. J Clin Med 2021;10:1514.
- Aune D, Schlesinger S, Norat T, Riboli E. Tobacco smoking and the risk of atrial fibrillation: A systematic review and meta-analysis of prospective studies. *Eur J Prev Cardiol* 2018;25:1437–1451.
- Cha MJ, Oh GC, Lee H, Park HE, Choi S-Y, Oh S. Alcohol consumption and risk of atrial fibrillation in asymptomatic healthy adults. *Heart Rhythm* 2020;**17**:2086–2092.
- Gillis AM. Atrial fibrillation and ventricular arrhythmias sex differences in electrophysiology, epidemiology, clinical presentation, and clinical outcomes. *Circulation* 2017;135:593–608.
- Kjerpeseth LJ, Igland J, Selmer R, Ellekjær H, Tveit A, Berge T, et al. Prevalence and incidence rates of atrial fibrillation in Norway 2004–2014. *Heart* 2021;107:201–207.
- Magnussen C, Niiranen TJ, Ojeda FM, Gianfagna F, Blankenberg S, Njølstad I, et al. Sex differences and similarities in atrial fibrillation epidemiology, risk factors, and mortality in community cohorts: results from the BiomarCaRE consortium (biomarker for cardiovascular risk assessment in Europe). *Circulation* 2017;**136**:1588–1597.
- Taubman SL, Robins JM, Mittleman MA, Hernán MA. Intervening on risk factors for coronary heart disease: an application of the parametric g-formula. Int J Epidemiol 2009;38:1599–1611.
- Robins JM, Hernan MA. Estimation of the casual effect of time-varying exposures. In: Fitzmaurice GM, Davidian M, Verbeke G, Molenberghs G, eds. *Longitudinal data analysis*. Boca Raton, FL: Chapman & Hall/CRC Press; 2008. p. 572–576.

- Wilsgaard T, Vangen-Lønne AM, Mathiesen E, Løchen M-L, Njølstad I, Heiss G, et al. Hypothetical interventions and risk of myocardial infarction in a general population: application of the parametric g-formula in a longitudinal cohort study-the Tromsø study. BMJ Open 2020;10:e035584.
- Danaei G, Robins JM, Young JG, Hu FB, Manson JE, Hernán MA. Weight loss and coronary heart disease: sensitivity analysis for unmeasured confounding by undiagnosed disease. *Epidemiology* 2016;**27**:302–310.
- Vangen-Lønne AM, Ueda P, Gulayin P, Wilsgaard T, Mathiesen EB, Danaei G. Hypothetical interventions to prevent stroke: an application of the parametric g-formula to a healthy middle-aged population. *Eur J Epidemiol* 2018;**33**:557–566.
- Jain P, Suemoto CK, Rexrode K, Manson JE, Robins JM, Hernán MA, et al. Hypothetical lifestyle strategies in middle-aged women and the long-term risk of stroke. Stroke 2020; 51:1381–1387.
- Danaei G, Pan A, Hu FB, Hernán MA. Hypothetical midlife interventions in women and risk of type 2 diabetes. *Epidemiology* 2013;24:122–128.
- Textor J, van der Zander B, Gilthorpe MS, Liskiewicz M, Ellison GT. Robust causal inference using directed acyclic graphs: the R package 'dagitty'. Int J Epidemiol 2016;45:1887–1894.
- Morseth B, Graff-Iversen S, Jacobsen BK, Jørgensen L, Nyrnes A, Thelle DS, et al. Physical activity, resting heart rate, and atrial fibrillation: the Tromsø study. Eur Heart J 2016;37:2307–2313.
- Lunde ED, Joensen AM, Lundbye-Christensen S, Fonager K, Paaske Johnsen S, Larsen ML, et al. Socioeconomic position and risk of atrial fibrillation: a nationwide Danish cohort study. J Epidemiol Community Health 2020;74:7–13.
- Kornej J, Börschel CS, Benjamin EJ, Schnabel RB. Epidemiology of atrial fibrillation in the 21st century. *Circ Res* 2020;**127**:4–20.
- 39. Statistics Norway. Statbank. https://www.ssb.no/en/statbank/ (23 September 2022)